Author's response to reviews

Title: The Impact of Regional Co-payment and National Reimbursement Criteria on Statins use in Italy. An Interrupted Time-Series Analysis.

Authors:

Gianfranco Damiani (gdamiani@rm.unicatt.it)
Bruno Federico (b.federico@unicas.it)
Angela Anselmi (angela_anselmi@libero.it)
Caterina BNA Bianchi (caterinabianchi@hotmail.it)
Giulia Silvestrini (silvestrinigiulia@gmail.com)
Lanfranco Iodice (lanfranco.iodice@gmail.com)
Pierfrancesco Navarra (pnavarra@rm.unicatt.it)
Roberto Da Cas (roberto.dacas@iss.it)
Roberto Raschetti (roberto.raschetti@iss.it)
Walter Ricciardi (wriccardi@rm.unicatt.it)

Version: 3 Date: 30 December 2013

Author's response to reviews:

Dear Editor,

We are glad for your intention of accepting our paper. According to your indication, some revisions were undertaken, listed as follows:

1) An ethical statement, along with a more specific source of data, was added in the “Methods” section, “Data” subparagraph, page 4: 3rd and 4th sentence.

2) The “Competing interests” section was slightly modified.

3) We added the “Acknowledgements” section at the end of the manuscript, just before the “References” section, also including a funding statement, according to the authors instructions of this Journal. Page 8.

4) A blank page (page 14) in the “Tables” section was deleted.